# UC San Diego UC San Diego Previously Published Works

# Title

Sitravatinib in patients with solid tumors selected by molecular alterations: results from a Phase Ib study.

**Permalink** https://escholarship.org/uc/item/08b3k43w

**Journal** Future Oncology, 20(39)

# Authors

Bazhenova, Lyudmila Kim, Dong-Wan Cho, Byoung <u>et al.</u>

# **Publication Date**

2024-12-01

# DOI

10.1080/14796694.2024.2418285

Peer reviewed

# RAPID COMMUNICATION

OPEN ACCESS Check for updates

# Sitravatinib in patients with solid tumors selected by molecular alterations: results from a Phase Ib study

Lyudmila Bazhenova<sup>\*,a</sup>, Dong-Wan Kim<sup>b</sup>, Byoung Chul Cho<sup>c</sup>, Sanjay Goel<sup>‡,d</sup>, Rebecca Heist<sup>e</sup>, Theresa L Werner<sup>f</sup>, Keith D Eaton<sup>g</sup>, Judy S Wang<sup>h</sup>, Shubham Pant<sup>i</sup>, Douglas R Adkins<sup>j</sup>, Collin M Blakely<sup>k</sup>, Xiaohong Yan<sup>§,I</sup>, Saskia Neuteboom<sup>#,I</sup>, James G Christensen<sup>#,I</sup>, Richard Chao<sup>I</sup> and Todd Bauer<sup>m</sup>

<sup>a</sup>University of California San Diego Moores Cancer Center, CA 92093, USA; <sup>b</sup>Seoul National University College of Medicine & Seoul National University Hospital, Seoul, Republic of Korea; <sup>c</sup>Yonsei Cancer Center, Yonsei University College of Medicine, Seoul, Republic of Korea; <sup>d</sup>Montefiore Medical Center, Bronx, NY 10467, USA; <sup>e</sup>Massachusetts General Hospital, Boston, MA 02114, USA; <sup>f</sup>Huntsman Cancer Institute, University of Utah, Salt Lake City, UT 84112, USA; <sup>g</sup>University of Washington & Fred Hutchinson Cancer Center, Seattle, WA 98109, USA; <sup>h</sup>Florida Cancer Specialists & Research Institute, Sarasota, FL 34232, USA; <sup>i</sup>University of Texas MD Anderson Cancer Center, Houston, TX 77030, USA; <sup>j</sup>Washington University School of Medicine in St Louis, MO 63110, USA; <sup>k</sup>University of California San Francisco, CA 94158, USA; <sup>I</sup>Mirati Therapeutics Inc., San Diego, CA 92121, USA; <sup>m</sup>Sarah Cannon Research Institute, Tennessee Oncology, Nashville, TN 37203, USA

### ABSTRACT

**Aim:** We report clinical activity and safety of sitravatinib in patients with advanced cancer from basket cohorts with specific molecular alterations, in a Phase Ib study.

**Materials & methods:** Patients with advanced solid tumors harboring amplification, mutation, or rearrangement of *MET*, *AXL*, *RET*, *NTRK*, *DDR2*, *KDR*, *PDGFRA*, *KIT* or *CBL* received sitravatinib once daily. Primary end point was confirmed objective response rate (ORR).

**Results:** In total, 113 patients were enrolled following a median of 3 (range 1–18) prior systemic regimens. Altered *RET* (n = 31), *CBL* (n = 31) and *MET* (n = 17) were most frequent cohorts. Overall, 68.9% had reduced tumor volume and most (61.5%) had a best objective response of stable disease. ORR was highest in patients with *RET*-rearranged non-small cell lung cancer (21.1%) but did not differ significantly from the null hypothesis (ORR  $\leq$ 15%; *p* = 0.316). Median progression-free survival and overall survival (5.7 and 24.2 months, respectively) were also longest in the *RET*-rearranged non-small cell lung cancer cohort. Diarrhea (61.1%), fatigue (50.4%) and hypertension (46.9%) were the most frequent treatment-emergent adverse events. Most treatment-emergent adverse events were mild-to-moderate in severity. The study closed before the planned number of patients were enrolled in all cohorts.

**Conclusion:** Sitravatinib had a manageable safety profile with modest signals of clinical activity in patients with molecularly selected solid tumors.

Clinical trial registration: www.clinicaltrials.gov identifier is NCT02219711

### **PLAIN LANGUAGE SUMMARY**

We report findings from a clinical study of sitravatinib which included patients with cancer that could not be removed by surgery or had spread to other parts of the body. The tumors of these patients contained specific molecular changes in one of the following genes: *MET*, *AXL*, *RET*, *NTRK*, *DDR2*, *KDR*, *PDGFRA*, *KIT* or *CBL*. All patients received treatment with sitravatinib once a day. Change in tumor size over time was assessed to see how effective treatment with sitravatinib was.

In total, 113 patients joined the study. Most patients had already received a median of three different types of medicines for their cancer (and up to 18 different types of anticancer medicines). Most patients had tumors that contained alterations in *RET*, *CBL* or *MET* genes.

During the study, the percentage of patients who had a decrease in the tumor size was highest in the group with non-small cell lung cancer that contained an altered *RET* gene (21.1%). However, this level of response to sitravatinib was not considered high enough to be medically important.

The most common side effects during the study were diarrhea (61.1%), fatigue (50.4%) and high blood pressure (46.9%). Most side effects were mild or moderate in severity. The study provided the

CONTACT Lyudmila Bazhenova 🖂 lbazhenova@health.ucsd.edu

B Supplemental data for this article can be accessed at https://doi.org/10.1080/14796694.2024.2418285

© 2024 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group

This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivatives License

(http://creativecommons.org/licenses/by-nc-nd/4.0/), which permits non-commercial re-use, distribution, and reproduction in any medium, provided the original work is properly cited, and is not altered, transformed, or built upon in any way. The terms on which this article has been published allow the posting of the Accepted Manuscript in a repository by the author(s) or with their consent.

### **ARTICLE HISTORY**

Received 26 July 2024 Accepted 15 October 2024

#### **KEYWORDS**

basket study; MGCD516; molecular alteration; sitravatinib; solid tumor



<sup>&</sup>lt;sup>‡</sup>Current affiliation: Rutgers Cancer Institute of New Jersey, New Brunswick, NJ, USA

<sup>&</sup>lt;sup>§</sup>Current affiliation: Alterome Therapeutics, San Diego, CA, USA

<sup>&</sup>lt;sup>#</sup>No longer employed at Mirati Therapeutics, San Diego, CA, USA

opportunity to assess sitravatinib as a treatment for cancers with specific gene mutations that are uncommon; the study closed before the planned number of patients were enrolled. In conclusion, the side effects seen in patients who received sitravatinib were manageable. Signals of how well sitravatinib worked were modest in patients with cancer that had spread to other parts of the body and contained specific molecular changes.

# **1. Introduction**

Receptor tyrosine kinases (RTK) play a key role in regulating numerous key cellular processes including proliferation, differentiation and migration through a variety of interconnected signaling pathways [1,2]. Molecular alterations such as gain-of-function mutations, genomic amplification and chromosomal rearrangement can lead to aberrant RTK downstream signaling that is not subject to normal 'checks and balances'. Indeed, constitutive RTK activation disrupts the balance between cell proliferation and death, and can trigger oncogenesis [1]. RTK inhibitors are central for the treatment of numerous cancer types with hallmark of dysregulated RTK signaling. However, apart from rare exceptions, cancer is not cured by treatment with a single RTK inhibitor, due in part to emerging resistance mechanisms [3]. Consequently, new treatment approaches are needed.

Sitravatinib (MGCD516) is an oral small molecule inhibitor that targets a spectrum of closely related RTKs implicated in oncogenesis, predominantly TAM family (TYRO3, AXL, MERTK) and split family (VEGFR2, MET, RET, KIT) receptors [4,5]. Sitravatinib demonstrated antiproliferative effects in several cancer cell lines and was a potent suppressor of tumor growth in xenograft models of tumors with RTK dysfunction [5–7]. The first-in-human study of sitravatinib evaluated pharmacokinetics (steady absorption supported once-daily administration) and dosing in patients with advanced solid tumors [4]. Safety and clinical activity were further assessed in Phase Ib expansion cohorts that enrolled patients with advanced, refractory tumors of selected histologic diagnoses (clear cell renal cell carcinoma or castrate-resistant prostate cancer with bone metastases) or molecular alterations relevant to the mechanism of sitravatinib; the latter utilized a basket study approach.

Use of basket trials in oncology settings has gained momentum. This approach is based on the drive for precision oncology to ensure patients receive treatment based on the molecular signature of their disease, and to overcome the recruitment challenges of prospective studies in settings of rare genetic alterations [8,9]. Simultaneously enrolling patients with a variety of tumor types containing specific target gene alterations enables signals of clinical activity to be identified as they are observed across multiple indications, with potential to further investigate signals of clinical activity in expansion cohorts [8]. While most basket trials are exploratory, this approach has resulted in approval of a limited number of cancer treatments. For example, larotrectinib was approved for solid tumors with *NTRK* gene fusions based in part on data from the SCOUT and NAVIGATE basket trials [10,11].

A basket-cohort approach was selected for the Phase Ib portion of the first-in-human sitravatinib clinical trial to facilitate enrollment of patients with tumors harboring molecular alterations relevant to the mechanism of action of sitravatinib. This included gene alterations in the targets of sitravatinib such as *MET*, *RET* and *AXL*, as well as amplification of chromosomal segment 4q12 (Chr4q12) which encodes several relevant oncogenic driver RTKs including *KIT*, *PDGFRA* and *KDR* [12]. Loss of function alterations in *CBL* were also included. *CBL* encodes E3ubiquitin ligase which facilitates the degradation of several RTKs implicated in carcinogenesis, including targets of sitravatinib [13–15]. Consequently, inactivation of *CBL* was postulated to lead to increased RTK density and signaling, thereby contributing to oncogenesis.

We report clinical activity and safety with sitravatinib in patients with solid tumors harboring genetic alterations relevant to the mechanism of action of sitravatinib, who participated in the Phase Ib basket study cohorts of the first-in-human study.

## 2. Materials & methods

## 2.1. Study design

This open-label, Phase I/Ib clinical trial (NCT02219711, 14 August 2014) included periods evaluating the pharmacokinetics (lead-in period), dosing (Phase I), and clinical activity (Phase Ib) of sitravatinib. Details of the study design have been reported [4]. Enrollment into the Phase Ib cohorts was based on a histologic diagnosis of renal cell carcinoma or castrate-resistant prostate cancer (reported separately) or by selected molecular alterations relevant for sitravatinib mechanism of action irrespective of histologic diagnosis (detailed below).

Patients in the Phase lb portion of the study received sitravatinib at the maximum-tolerated dose established in the Phase I cohort: 150 mg/day [4]. During the study, the starting dose was reduced to 120 mg/day following evaluation of cumulative safety and tolerability data.

Dose reductions and interruptions were permitted for adverse events (AEs) assessed by the study investigators as related to study medication, and study treatment was continued at the discretion of the investigator until disease progression, unacceptable toxicity or withdrawal of consent.

# 2.2. Study population

Eligible patients were  $\geq$ 18 years and had advanced, unresectable, or metastatic solid tumors for which standard treatment was not available. Patients also had a life expectancy of  $\geq$ 3 months and Eastern Cooperative Oncology Group (ECOG) performance status 0-2. There were no restrictions on the number of prior lines of therapy, and prior treatment with specific therapies targeting molecular markers of interest were permitted on a case-by-case basis. Patients had not received anticancer therapy for  $\geq 2$  weeks prior to their first dose of study treatment and had recovered from any AEs to baseline or Grade 1 (except for alopecia). Patients were excluded with unacceptable hepatic, renal and bone marrow function, symptomatic or uncontrolled brain metastases, or significant cardiac abnormalities within the prior 6 months. Other exclusion criteria included prolonged QTc interval (>480 ms), left ventricular ejection fraction <40%, uncontrolled arterial hypertension, another active cancer (excluding basal cell carcinoma or cervical intraepithelial neoplasia), and major surgery  $\leq 4$  weeks before the first dose of study medication.

Patients with tumors harboring amplification, mutation, or rearrangement of *MET*, *AXL*, *RET*, *NTRK*, *DDR2*, *KDR*, *PDGFRA*, *KIT* or *CBL* were enrolled into an overall basket cohort as they were identified (amplifications of *MET*, *Chr4q12* and *AXL* were defined as *MET*: chromosome 7 centromere [CEP7] ratio  $\geq$ 5:1, *KIT*:CEP7 ratio of 5:1, and *AXL*  $\geq$ 8 copies, respectively). Molecular alterations were identified in tumor tissue or ctDNA using quantitative RT-PCR (*MET* exon 14 skipping mutations), fluorescence *in situ* hybridization (*RET* rearrangements and *MET* amplification) and next-generation sequencing (any genetic alterations). Tumor samples were required for retrospective central laboratory confirmation if molecular eligibility was established locally.

While sufficient patients with *RET* alterations were required to evaluate the clinical activity of sitravatinib, the feasibility of enrolling enough individuals with other relatively rare gene alterations into potential dedicated molecular cohorts within a reasonable timeframe was taken into consideration. Populations of interest emerged from the basket cohort as clinical activity signals were observed in clusters of patients with unifying targeted gene alterations (with or without a specific histologic diagnosis).

# 2.3. Study objectives & assessments

The primary objective in the Phase Ib molecular cohorts was to assess the clinical activity and safety of sitravatinib. Confirmed objective response rate (ORR), the primary efficacy end point, was assessed in accordance with Response Evaluation Criteria in Solid Tumors (RECIST) v1.1. Scans (computed tomography or magnetic resonance imaging) of known or suspected disease sites were obtained at baseline and at 6-week intervals during the study. Duration of response, progressionfree survival (PFS) and overall survival (OS) were also assessed. Treatment-emergent adverse events (TEAEs) were graded per National Cancer Institute Common Terminology Criteria for Adverse Events (NCI CTCAE) v4.03. Additional safety assessments included clinical laboratory parameters, physical examinations, vital sign, electrocardiogram and left ventricular ejection fraction measurements.

# 2.4. Statistical analysis

A two-stage Simon optimal design was used to identify specific molecular alterations for further study. If an objective response was seen in  $\geq 2$  of eight patients with the same molecular alteration (or a subgroup with the same histologic diagnosis and molecular alteration), an additional 16 patients were enrolled. Further cohort expansion was permitted if objective responses were seen in  $\geq 7$  of these 24 patients.

Summaries of ORR and corresponding 95% CI were calculated using the binomial proportions confidence interval method. Exact test for single proportion (one-sided alpha = 2.5%) tested the alternative hypothesis of ORR >15% (null hypothesis of ORR  $\leq$ 15%).

Duration of response (time from first documentation of completed response [CR] or partial response [PR]) to disease progression [PD] per RECIST v1.1, or death due to any cause), PFS (time from first dose of study medication to PD or death due to any cause]) and OS (time from first dose of study medication to death due to any cause) were estimated using Kaplan–Meier methodology. Other data were summarized using descriptive statistics.

Response was assessed in the clinical activity evaluable population which included patients who received  $\geq 1$ cycle of therapy at  $\geq 80\%$  of assigned dose and had  $\geq 1$  on-study disease assessment. Other clinical activity assessments and safety were evaluated in the modified intent-to-treat population which comprised patients who received  $\geq 1$  dose of study medication. The primary data cut-off was 31 July 2020. Data from patients who remained on treatment (n = 2 in *AXL* cohort) and long-term follow-up are included up to 10 October 2022.

# 3. Results

# 3.1. Patient characteristics

Genetic testing data were available for 639 of 734 patients screened between 27 August 2014 and 5 February 2020. Overall, 113 of 152 patients with qualifying genetic alterations were enrolled into distinct molecular cohorts.

Thirty-one patients had RET alterations in any tumor type. This cohort included a subgroup with non-small cell lung cancer (NSCLC) and any *RET* rearrangement (n = 23) which included fusion partners KIF5B (n = 13), CCDC6 (n = 3), DSP (n = 1) and not specified (n = 6). Thirtyone patients were also enrolled into the CBL alteration cohort which included missense mutations (n = 25), indel (n = 5), or splice site mutation (n = 1) resulting in CBL inactivation. The MET altered cohort included patients (n = 17) with tumors harboring MET exon 14 skipping (n = 9), MET amplification (n = 4), MET point mutations D1246H (n = 1), R988C (n = 1), R988C with *MET* amplification (n = 1) and MET overexpression (n = 1). Sixteen and seven participants were enrolled with amplification of chromosome segment 4q (Chr4q) and AXL amplification, respectively. The cohort with other molecular alterations (n = 11) included changes that involved KIT (n = 5), KDR (n = 3), NTRK (n = 2), or DDR2 (n = 1); one patient with both Chr4q amplification and KIT alteration was included in the Chr4q amplification cohort.

Across the cohorts, patients had a median age of 66 (range 36–87) years and NSCLC was the most frequent histologic diagnosis (46.9%). In general, participants were heavily pretreated, having received a median of 3 (range 1–18) prior systemic regimens and over half had received prior radiotherapy (54.0%). Baseline demographic and disease characteristics were broadly balanced across the molecular alteration cohorts (Table 1). Prior RET inhibitor treatment was reported in three patients with medullary (neuroendocrine) thyroid cancer and prior MET inhibitor treatment was reported in two patients with NSCLC in the *RET* alteration and *MET* alteration cohorts, respectively.

Across all patients, the most frequent reasons for study discontinuation were death (71.7%, n = 81) and withdrawal of consent (16.8%, n = 19). Objective disease progression (50.4%, n = 57), AEs (16.8%, n = 19) and withdrawal of consent (14.2%, n = 16) were the most frequent reasons for discontinuing study treatment. Reasons for

discontinuation from the study and study treatment were balanced across the molecular cohorts (Figure 1).

# 3.2. Antitumor activity

Across the molecular alteration cohorts, 68.9% (51 of 74 with evaluable data) of patients experienced reductions in tumor volume, which were particularly pronounced in some individuals with RET alterations (Figure 2). Confirmed ORRs did not differ significantly from the null hypothesis in any cohort (Table 2). In the RET altered cohort, ORR was 19.2% (five of 26 evaluable patients achieved PR). Four of five PRs were seen in patients with NSCLC and RET rearrangement (lung adenocarcinoma with RET alteration not specified [n = 2], CCDC6-RET rearrangement [n = 1], *KIF5B-RET* rearrangement [n = 1]) and one PR occurred in a patient with hormone receptor (HR)-positive breast adenocarcinoma with RET C634R. Duration of responses ranged from 1.8 to 10.2 months with a Kaplan-Meier estimate of 40.0% (95% CI: 5.2-75.3) for ongoing response at 6 months. Maximum reported change in target lesion ranged from -34.1 to -100% in responders (Figure 2A). SD occurred in 61.5% (16/26 evaluable patients), including one patient with unconfirmed response and maximum target lesion change of -36.7%. Of note, SD (maximum target lesion change -24.5% lasting 5.3 months) was reported in one of three patients enrolled with medullary or neuroendocrine thyroid cancer and RET activating mutations, who were previously treated with >2 RET inhibitors.

In the *MET* alteration cohort confirmed ORR was 15.4% with PR reported in 2/13 evaluable patients (NSCLC with MET overexpression [n = 1] and *MET* exon 14 skipping [n = 1]). While nine of 11 evaluable patients experienced some reduction in tumor volume, maximum target lesion change was -41.6 and -49.4% in the responders (Figure 2B). Duration of response was 3.0 months in one patient and the second was censored at 5.6 months. Most patients in the *MET* alteration cohort achieved SD (76.9%, n = 10/13 evaluable patients) including n = 2 (both with *MET* exon 14 skipping alterations) with unconfirmed response and maximum target lesion changes of -34.8 and -50.0%.

There were no objective responses in the Chr4q12 amplification cohort, with most patients having SD (81.8%; 9/11 evaluable patients [Figure 2C]). In 18 evaluable patients with *CBL* alteration 44.4% (n = 8) had SD and the was one PR (ORR 5.6%) which lasted 14.2 months and occurred in an individual with sinonasal melanoma and *CBL* Y368C. Maximum target lesion change was -50.5% (Figure 2D). One additional PR which lasted 4.3 months (maximum target lesion change -77.4%) was observed in a patient with NSCLC and *CBL* C384R who received

	טוור מווח מוזכמזב	רוומומרובווזיוורז חו ווו	ב וווחוברמומו מוובומ		זווובת ווונבוויר-יה-יובנ	ar population).			
n (%)	All patients (N = 113)		<i>RET</i> alteration		<i>MET</i> alteration (N = 17)	Chr4q12 amplification <sup>b</sup> (N = 16)	<i>CBL</i> alteration (N = 31)	AXL amplification $(N = 7)$	Other $(N = 11)$
		Overall <i>RET</i> cohort (N = 31)	NSCLC and <i>RET</i> re	earrangement <sup>a</sup>					
			All patients (N = 23)	KIF5B-RET (N = 13)					
Median age	66 (36–87)	62 (45–81)	62 (45–71)	58 (46–71)	70 (36–81)	67 (50–77)	69 (44–87)	65 (43–74)	68 (49–79)
(range), years Male	47 (41.6)	12 (38.7)	9 (39.1)	4 (30.8)	6 (35.3)	7 (43.8)	14 (45.2)	3 (42.9)	5 (45.5)
etnnicity White	78 (69.0)	16 (51.6)	10 (43.5)	4 (30.8)	13 (76.5)	9 (56.3)	25 (80.6)	6 (85.7)	9 (81.8))
African American	8 (7.1)	2 (6.5)	2 (8.7)	1 (7.7)	2 (11.8)	3 (18.8)	1 (3.2)	0	0
Asian Other	21 (18.6) 6 (5.3)	10 (32.3) 3 (9.7)	9 (39.1) 2 (8.7)	6 (46.2) 2 (15.4)	2 (11.8) 0	4 (25.0) 0	3 (9.7) 2 (6.5)	1 (14.3) 0	1 (9.1) 1 (9.1)
Primary diagnosis Head and neck	2 (1.8)	0	0	0	0	1 (6.3)	0	1 (14.3)	0
cancer Hormone- refractory	3 (2.7)	0	0	0	1 (5.9)	1 (6.3)	1 (3.2)	0	0
prostate cancer									
NSCLC	53 (46.9)	24 (77.4)	23 (100)	13 (100)	12 (70.6) <sup>c</sup>	4 (25.9)	10 (32.3)	1 (14.3)	2 (18.2)
Breast cancer Colon/rectal	4 (3.5) 7 (6.2)	2 (6.5) 1 (3.2)	0 0	0 0	0 2 (11.8)	1 (6.3) 0	1 (3.2) <sup>e</sup> 3 (9.7)	0 0	0 1 (9.1)
cancer									
Melanoma Soft tissue	10 (8.8) 7 (6.2)	0 0	0 0	0 0	0 0	3 (18.8) 1 (6.3)	5 (16.1) 4 (12.9) <sup>e</sup>	0 0	2 (18.2) 2 (18.2)
sarcoma									
Other ECOG	29 (25.7)	4 (12.9) <sup>d</sup>	0	0	3 (17.6)	5 (31.3)	8 (2.6)	5 (71.4)	4 (36.4)
performance score									
0	33 (29.2)	9 (29.0)	8 (34.8)	5 (38.5)	2 (11.8)	3 (18.8)	11 (35.5)	4 (57.1)	4 (36.4)
1	70 (61.9)	17 (54.8)	11 (47.8)	6 (46.2)	15 (88.2)	12 (75.0)	16 (51.6)	3 (42.9)	7 (63.6)
2	10 (8.8)	5 (16.1)	4 (17.4)	2 (15.4)	0	1 (6.3)	4 (12.9)	0	0
Prior surgery Prior	70 (61.9) 61 (54 0)	16 (51.6) 21 (67 7)	10 (43.5) 14 (60 9)	7 (53.8) 9 (69 2)	8 (47.1) 9 (52 9)	10 (62.5) 13 (81 3)	20 (64.5) 14 (45 2)	7 (100) 1 (14 3)	9 (81.8) 3 (77 3)
radiotherapy									
Prior systemic	101 (89.4)	30 (96.8)	22 (95.7)	13 (100)	14 (82.4)	15 (93.8)	26 (83.9)	5 (71.4)	11 (100)
Median (range) number of prior	3 (1–18)	3 (1–11) <sup>f</sup>	2 (1–6)	3 (1–4)	3 (1–9) <sup>g</sup>	3 (1–18)	3 (1–6)	4 (2–5)	3 (1–7)
regimens									

Table 1. Domocraphic and discoses characteristics of the molecular alteration cohorts (modified intent to treat nonulation)

<sup>a</sup> Includes patients with NSCLC and *RET* fusion partner *KIF5B, CCDC6, DSP* or not specified. <sup>b</sup>Includes one patient with Chr4q12 amplification and *KIT* alteration. <sup>c</sup>One patient had concurrent NSCLC and renal cell carcinoma. <sup>d</sup>Includes three patients with thyroid cancer. <sup>e</sup>One patient had two primary diagnoses (soft tissue sarcoma and breast cancer). <sup>f</sup>Prior treatment with = 2 RET inhibitors (vandetanib, cabozantinib and lenvatinib) was reported in three patients with thyroid cancer. <sup>9</sup>Prior treatment with a MET inhibitor (crizotinib and/or salvolitinib) was reported in two patients with NSCLC. ECOG: Eastern Cooperative Oncology Group; NSCLC: Non-small cell lung cancer.

FUTURE ONCOLOGY

3217



Figure 1. Disposition of patients enrolled in the molecular alteration basket cohorts (modified intent-to-treat population). <sup>a</sup>Subgroup of 'Overall *RET* alteration' cohort;

<sup>b</sup>Subgroup of 'Overall RET alteration' and 'NSCLC RET rearrangement' cohorts.

a total dose <80% in Cycle 1; while this patient did not qualify for the clinical activity evaluable population, they received sufficient study treatment overall to be considered clinically evaluable.

In the AXL amplification cohort, PR lasting 3.1 months was seen in 1/6 evaluable patients (ORR 16.7%, Table 2) and occurred in a patient with bladder adenocarcinoma who had a maximum target lesion change of -32.1% (Figure 2E). Most patients (66.7%) had SD, including an individual with NSCLC who had achieved PR of approximately 3.6 years in a prior glesatinib clinical trial and was progression free at study completion. The patient then enrolled into the current study with two non-target lung lesions and remained progression free at last on-study assessment (prolonged SD >2 years 1 month [followed for 776 days]) and continues to be progression free following enrolment into an ongoing sitravatinib rollover study (2.4 years; ongoing disease control for  $\sim$ 8 years). In the cohort comprising patients with other relevant molecular alterations ORR was 12.5% (Table 2). One of eight evaluable patients (thymic carcinoma with KIT V560 deletion) had a PR that lasted for 15.2 months and a

maximum target lesion change of -48.6% (Figure 2F). Most patients (62.5%) achieved SD.

Median PFS was 5.7 months in patients with NSCLC harboring RET rearrangement, and specifically KIF5B-RET, and 6-month PFS estimates were 40.3 and 36.4%, respectively (Figure 3A). Median PFS and 6-month PFS estimates were shorter in the Chr4q12 amplification, MET and CBL alteration cohorts, ranging from 2.0-2.9 months and 12.3-23.8%, respectively (Figure 3A). While median PFS was not reported for the AXL amplification cohort, 6month PFS estimate was the longest of the molecular alteration cohorts (71.4%). Two patients in the AXL amplification cohort who continued study treatment following primary data cutoff had PFS of 351 days and censored PFS of 776 days (patient was progression-free at last recorded follow-up, described above). The longest duration of OS was seen in patients with NSCLC harboring RET rearrangement, including the KIF5B-RET subgroup (median OS 24.2 months in both cohorts; 12-month OS estimate 73.2 and 79.5%, respectively). In patients with tumors harboring Chr4q12 amplification and MET, CBL or AXL alterations, median OS and 12-month OS



**Figure 2.** Percentage change in tumor burden grouped by molecular alteration (clinical activity evaluable population<sup>a</sup>). **(A)** *RET* alteration (N = 23)<sup>b</sup>. **(B)** *MET* alteration (N = 11). **(C)** Chr4q12 amplification (N = 11). **(D)** *CBL* alteration (N = 16)<sup>c</sup>. **(E)** *AXL* amplification (N = 5). **(F)** Other alteration (N = 8).

<sup>a</sup>Patients were excluded with missing data for change from baseline.

<sup>b</sup>Overall *RET* alteration cohort is shown.

<sup>c</sup>One additional PR (maximum target lesion change -77.4%) was observed in a patient with NSCLC and *CBL* C384R who did not qualify for the CAE population (total dose of sitravatinib <80% in Cycle 1) but received sufficient study treatment overall to be considered clinically evaluable.

CAE: Clinical activity evaluable; NSCLC: Non-small cell lung cancer; PD: Disease progression; PR: Partial response; RECIST: Response Evaluation Criteria in Solid Tumors; SD: Stable disease.

estimate ranged from 5.0–9.5 months and 20.2–35.2%, respectively (Figure 3B).

# 3.3. Study treatment exposure

Across the molecular alteration cohorts the starting dose of sitravatinib was 150 mg QD and 120 mg QD in n = 60 and n = 53, respectively. Patients started a median of 4.0 cycles of study treatment and relative dose intensity was 80.5%. The two patients in the *AXL* alteration cohort

who continued study treatment following the primary data cutoff started 39 cycles and 16 cycles of sitravatinib (relative dose intensity was 92.1 and 73.8%, respectively). Exposure to sitravatinib across the molecular alteration cohorts was broadly similar (Supplementary Table 1).

# 3.4. Safety

Across all the molecular alteration cohorts, the most frequent all-cause TEAEs were diarrhea (61.1%

	-	-				-			1
(%) u		<i>RET</i> alteration		<i>MET</i> alteration (N = 13)	Chr4q12 amplification (N = 11)	CBL alteration (N = 18)	AXL amplification (N = 6)	Other $(N = 8)$	
	Overall $RET$ cohort (N = 26)	NSCLC <i>RET</i> rea	rrangement <sup>a</sup>						
		All patients ( $N = 19$ )	KIF5B-RET (N = 9)						
Best overall									
response, n (%)									
ORR	5 (19.2)	4 (21.1)	1 (11.1)	2 (15.4)	0	1 (5.6)	1 (16.7)	1 (12.5)	
95% CI	6.6–39.4	6.1-45.6	0.3-48.2	9.1, 45.4	0.0-28.5	0.1–27.3	0.4–64.1		
P value <sup>b</sup>	0.350	0.316	0.768	0.602	>0.999	0.946	0.623		
CR	0	0	0	0	0	0	0	0	
PR	5 (19.2)	4 (21.1)	1 (11.1)	2 (15.4)	0	1 (5.6) <sup>c</sup>	1 (16.7)	1 (12.5)	
SD	16 (61.5)	12 (63.2)	5 (55.6)	10 (76.9)	9 (81.8)	8 (44.4)	4 (66.7)	5 (62.5)	
PD	5 (19.2)	3 (15.8)	3 (33.3)	1 (7.7)	2 (18.2)	8 (44.4)	1 (16.7)	2 (25.0)	
NE	0	0	0	0	0	1 (5.6)	0	0	
Median DoR,	2.8	2.3	10.2	N/A <sup>d</sup>	N/A	N/A <sup>e</sup>	3.1 <sup>f</sup>	15.2	
months									
(range)	(1.8, 10.2)	(1.8, 10.2)	(10.2, 10.2)	(3.0, 5.6)	N/A	(14.2, 14.2)	(3.1, 3.1)	(15.2, 15.2)	
<sup>a</sup> Includes patients wit	h NSCLC and <i>RET</i> fusion	partner KIF5B, CCDC6, DSP	or not specified.						
<sup>c</sup> One additional patie	nt with limited dosing in	Cycle 1 (did not qualify fo	r CAE population) also h	ad PR (not shown).					
<sup>d</sup> One patient experier	iced PD at 3.0 months ai	nd the second was censore	d at 5.6 months.						
<sup>e</sup> Duration of response	was 14.2 months (dura)	tion of response was 4.3 m	onths in the additional p	batient with PR who did I	not qualify for the CAE p	opulation).			
<sup>f</sup> PR ongoing at study	withdrawal.								

Table 2. Best overall response with sitravatinib in patients with tumors harboring molecular alterations (clinical activity evaluable population).

CAE: Clinical activity evaluable; CR: Complete response; DoR: Duration of response; N/A: Not applicable; NSCLC: Non-small cell lung cancer; PD: Disease progression; PR: Partial response; SD: Stable disease.

Chr4q12 amplification AXL amplification All molecular alterations

RET alteration MET alteration

CBL alteration





		0						
Number of patients at risk								•
All molecular alterations	113	50	23	9	7	5	3	0
Chr4q12 amplification	16	5	3	1	0			
AXL amplification	7	5	1	0				
RET alteration	31	18	11	5	5	3	2	0
MET alteration	17	7	2	1	0			
CBL alteration	31	10	3	1	1	1	0	

DE0.0		RET alteration			MET	Chr4a10	CBI	AYI
PFS Summary	All patients (N = 113)	Overall	NSCL rearran	C RET gement <sup>a</sup>	alteration	amplification	alteration	amplification
n (%)		(N = 31)	All patients (N = 23)	<i>KIF5B-RET</i> (N = 13)	(N = 17)	(N = 16)	(N = 31)	(N = 7)
Median PFS, months	3.6	5.3	5.7	5.7	2.9	2.7	2.0	NR
(95% CI)	(2.7, 4.4)	(3.0, 6.7)	(3.0, 13.5)	(1.5, 13.5)	(1.9, 4.3)	(1.3, 4.4)	(1.3, 4.1)	(1.2, NE)
6-month PFS KM estimate, %	28.5	39.5	40.3	36.4	16.4	23.8	12.3	71.4
(95% CI)	(19.6, 38.0)	(21.8, 56.7)	(19.5, 60.4)	(11.2, 62.7)	(2.8, 40.0)	(5.8, 48.5)	(3.1, 28.3)	(25.8, 92.0)





			RET alteration			Chr4q12 amplification (N = 16)	0.51	AVI
OS Summary	All patients (N = 113)	Overall RET schort	NSCL rearran	C RET gement <sup>a</sup>	alteration		CBL alteration	AXL amplification
n (%)		(N = 31)	All patients	KIF5B-RET	(N = 17)		(N = 31)	(N = 7)
		(14 = 51)	(N = 23)	(N = 13)				
Median OS, months	9.5	18.2	24.2	24.2	7.3	9.5	5.0	9.5
(95% CI)	(7.1, 10.9)	(10.9, 33.8)	(10.9, NE)	(10.9, NE)	(3.5, 13.9)	(3.6, 18.1)	(3.2, 9.9)	(2.2, NE)
12-month OS KM estimate, %	35.8	66.5	73.2	79.5	27.1	35.2	20.2	28.6
(95% CI)	(26.2, 45.6)	(43.3, 82.0)	(46.6, 88.1)	(39.3, 94.5)	(8.4, 50.2)	(13.3, 58.2)	(7.4, 37.3)	(4.1, 61.2)

Figure 3. Progression-free survival and overall survival with sitravatinib in patients with tumors harboring molecular alterations (modified intent-to-treat population).

(A) PFS. (B) OS.

<sup>a</sup>Includes patients with NSCLC and RET fusion partner KIF5B, CCDC6, DSP or not specified.

CI: Confidence interval; KM: Kaplan–Meier; NE: not evaluable; NR: not reported; NSCLC: Non-small cell lung cancer; OS: Overall survival; PFS: Progression-free survival.

[n = 69], fatigue (50.4% [n = 57]), hypertension (46.9% [n = 53]) and nausea (38.9% [n = 44]; Table 4). These events were frequently considered by the study investigators to be related to study treatment (diarrhea 54.0% [n = 61], fatigue 43.4% [n = 49], hypertension 42.5% [n = 48] and nausea 31.0% [n = 35]). Nausea and hypertension were also the most frequent serious AEs considered by the investigators to be related to study medication (both n = 3 [2.7%]); other treatment-related serious AEs were reported in one or two patients only. Most AEs were mild-to moderate in severity. Except for hypertension (n = 31 [27.4%]), fatigue (n = 12 [10.6%]) and diarrhea (n = 11 [9.7%]), Grade >3 TEAEs were reported in fewer than 6% of patients (Table 3). There were no marked differences in the incidence of treatment-related TEAEs across the molecular alteration cohorts.

Dose reductions or interruptions due to AEs were reported in 71.7% (n = 81), and discontinuation of sitravatinib due to AEs was reported in 20.4% (n = 23). All-cause AEs resulting in study treatment discontinuation included alanine aminotransferase increased, aspartate aminotransferase increased, diarrhea, hypertension and sepsis (n = 2, each). Disease progression was also reported as an all-cause AE resulting in sitravatinib discontinuation in three patients (other AEs were reported as single events).

Across the molecular alteration cohorts, 14 patients died during the study. Most deaths were due to disease progression (n = 11), and two and one patient died of sepsis and pneumonia, respectively.

# 4. Discussion

We report the clinical activity and safety of sitravatinib, an inhibitor of several oncogenic RTKs, including split and TAM family members, in a subset of a Phase Ib population who were enrolled using a basket approach. These patients had malignancies with molecular alterations relevant to the mechanism of action of sitravatinib. This included amplification, mutation, or rearrangement of sitravatinib molecular targets: MET, AXL, RET, NTRK, DDR2, KDR, PDGFRA and KIT. Patients with tumors harboring amplification of Chr4q12 were also enrolled as this genetic segment encodes several relevant oncogenic driver RTKs including KIT, PDGFRA and KDR [12]. Furthermore, durable clinical benefit has been reported in some patients with Chr4q12 amplified tumors who received other TKIs with known anti-PDGFRA and anti-KIT activity [12]. Patients were also enrolled with tumors containing loss-of-function alterations in CBL. The resulting loss of E3-ubiquitin ligase may result in decreased

degradation of several RTKs implicated in carcinogenesis, including targets of sitravatinib, potentially leading to increased RTK density and signaling, thereby contributing to oncogenesis [13–15].

Sitravatinib was previously reported to have modest clinical activity (ORR 11.8%) across all evaluable patients in this Phase I/Ib study, which also including those enrolled into cohorts based on histologic diagnosis alone [4]. However, the basket approach for the molecular alteration cohorts was designed to have flexibility to facilitate identification of patients who are most likely to benefit from treatment with sitravatinib. The molecular alteration cohorts could evolve from a broad population with regards to alteration for a given gene to become more refined (e.g., all tumor types with RET rearrangement to NSCLC with KIF5B-RET rearrangement), with the same Simon two-stage design used for the parent and refined cohorts. However, while clinical activity was seen in several molecular alteration cohorts, their low frequency combined with variability in the type of alteration within a specific gene and differing tumor types limited the feasibility to complete enrollment in all cohorts within a relevant timescale, and the trial was closed.

Across the molecular alteration cohorts, clinical activity signals with sitravatinib were most pronounced in patients with previously treated RET altered tumors, although the response rate did not differ significantly from the null hypothesis. Most patients experienced reductions in tumor volume, including PRs in four individuals with NSCLC harboring RET rearrangements at several different loci (confirmed ORR 21.1%) and in one patient with RET C634R-mutated HR-positive breast cancer; of note response duration was  $\geq$ 7 months in two of 26 evaluable patients with previously treated disease. Clinical meaningful disease control (SD lasting 5.3 months) was also observed in one of three patients with thyroid cancer harboring RET rearrangement. Of note, all three patients with thyroid cancer had received prior treatment with two or three non selective RET inhibitors (vandetanib, cabozantinib and lenvatinib) which have potential to evoke RET-targeted resistance mechanisms.

During the course of our study ORRs of 57–64% and 89–100% were reported in patients with *RET* fusion-positive NSCLC and *RET* fusion-positive thyroid cancer, respectively, across clinical trials of the RTK inhibitors, selpercatinib and pralsetinib [16,17]. These findings resulted in FDA approval of both drugs in these settings [16,17]. While direct comparison of outcomes across studies is not recommended due to differences in study design and patient populations, the ORR with selpercatinib and pralsetinib in patients with *RET*-altered NSCLC and thyroid cancer exceeds the modest, preliminary

Table 3. Frequent all-cause treatment-emergent adverse events ( $\geq$ 10% of patients) in study participants with tumors harboring molecular alterations (modified intent-to-treat population).

n (%)		All patients (N = 113)					
_	All Grade	Grade 3	Grade 4				
Gastrointestinal disorders							
Diarrhea	69 (61.1)	11 (9.7)	0				
Nausea	44 (38.9)	1 (0.9)	0				
Constipation	40 (35.4)	0	0				
Vomiting	31 (27.4)	3 (2.7)	0				
Abdominal pain	26 (23.0)	3 (2.7)	0				
Dry mouth	19 (16.8)	0	0				
Stomatitis	19 (16.8)	2 (1.8)	0				
Oral pain	13 (11.5)	0	0				
General disorders/administration							
site conditions							
Fatigue	57 (50.4)	12 (10.6)	0				
Peripheral edema	16 (14.2)	0	0				
Asthenia	12 (10.6)	0	0				
Metabolism and nutrition							
disorders							
Decreased appetite	43 (38.1)	3 (2.7)	0				
Dehydration	20 (17.7)	1 (0.9)	0				
Hyponatremia	15 (13.3)	6 (5.3)	1 (0.9)				
Hypokalemia	13 (11.5)	2 (1.8)	0				
Hypophosphatemia	13 (11.5)	6 (5.3)	0				
Respiratory, thoracic and							
mediastinal disorders							
Dysphonia	30 (26.5)	0	0				
Dyspnea	18 (15.9)	1 (0.9)	0				
Cough	15 (13.3)	0	0				
Vascular disorders		Ū.	Ū.				
Hypertension	53 (46 9)	31 (27 4)	0				
Investigations	33 (10.3)	51 (27.1)	Ũ				
AIT increased	30 (26.5)	4 (3.5)	0				
AST increased	29 (25.7)	1 (0.9)	0				
Weight decreased	27 (23.9)	2 (1.8)	0				
Lipase increased	14 (12 4)	4 (3 5)	2 (1.8)				
Skin and subcutaneous tissue	()	. (0.0)	2 (110)				
disorders							
Hand-foot syndrome	21 (18.6)	6 (5 3)	0				
Bash	16 (14.2)	2 (1.8)	0				
Nervous system disorders		2 (110)	Ū.				
Dizziness	23 (20.4)	0	0				
Headache	23 (20.4)	2 (1 8)	0				
Musculoskeletal and connective	25 (2011)	2 (110)	Ū.				
tissue disorders							
Back pain	16 (14.2)	0	0				
Pain in extremity	15 (13.3)	1 (0.9)	0				
Arthralgia	13 (11.5)	1 (0.9)	0				
Infections and infestations		. (0.2)	Ū.				
Urinary tract infection	13 (11.5)	3 (2.7)	0				
Endocrine disorders		0 (217)	Ū.				
Hypothyroidism	28 (24 8)	0	0				
Blood and lymphatic system	20 (2 1.0)	ő	5				
disorders							
Anemia	19 (16 8)	6 (5 3)	0				
Renal and urinary disorders			-				
Proteinuria	14 (12.4)	1 (0.9)	0				
	(. = /	. (0.2)					

ALT: Alanine aminotransferase; AST: Aspartate aminotransferase.

clinical activity observed with sitravatinib in these cohorts in our study. While durable responses were seen with sitravatinib in two patients with *RET*-altered tumors, additional refinement of predictive indicators would be required to warrant further investigation of sitravatinib in this setting. While this study was closed prior to enrollment of the n = 24 planned patients eligible for clinical activity in the other molecular cohorts, some signals of clinical activity were seen, including PRs in 2 of 18 evaluable patients with *CBL* missense mutations. A prolonged PR (14.2 months) was reported in a patient with sinonasal

melanoma and CBL Y368C that was resistant to several prior treatments, leading to the hypothesis that inhibiting PDGFR $\alpha$  and/or KIT may result in clinical activity in this setting. Another PR of clinically meaningful duration (4.3 months) occurred in a patient with NSCLC and CBL C384R with prior EGFR inhibitor failure, leading to the hypothesis that CBL mutation may result in resistance to EGFR inhibition through increased MET activity. Of note, a large proportion (11/39 [39%]) of patients with CBL altered tumors were not eligible for clinical activity evaluation, largely due to on-study death and sitravatinib dose modifications, suggesting CBL inactivation may be associated with poor prognosis. These observations suggest further refinement of CBL alteration type would be required to warrant future study of sitravatinib in patients with tumors harboring this molecular alteration.

Most evaluable patients with tumors harboring altered MET experienced reductions in tumor volume, including confirmed responses in two of 13 (15.4%) patients, both of whom had NSCLC, one harboring MET overexpression and one with MET exon 14 skipping. Since the onset of this study, capmatinib and tepotinib received FDA approval for patients with NSCLC and MET exon 14 skipping alterations, with ORRs of 45-68% and 41-45% in treatmentnaive and previously treated patients, respectively [18-20]. While the clinical activity of sitravatinib in METaltered NSCLC appears lower than that of capmatinib and tepotinib, both approved MET inhibitors are associated with toxicities that can be challenging in some patients including edema, pulmonary symptoms and hepatotoxicity [18]. This underscores the challenges of developing effective treatments that are well tolerated in this setting. Clinical activity signals were also observed in tumors harboring AXL amplification, with five of six evaluable patients having disease control (n = 1 PR and n = 4 SD), including one patient with NSCLC who achieved ongoing disease control for approximately 8 years with glesatinib (in a prior study) followed by sitravatinib. Given the roles of AXL in cell proliferation, survival, migration, regulation of natural killer cell development, and drug resistance mechanisms, along with the lack of approved treatments in this setting [21], further investigation of sitravatinib may be warranted for patients with AXL-altered tumors.

In this study, the safety assessment of sitravatinib showed gastrointestinal events were among the most frequent TEAEs considered by the investigators to be related to study treatment (diarrhea 54%, nausea, 31%) along with hypertension (42.5%) and fatigue (43.4%). These findings are aligned with safety observations previously reported across all patients enrolled in this Phase I/lb study, as well as in smaller studies of sitravatinib in combination with nivolumab [4,22–24]. No safety signals were identified that would impact further development of this investigational agent. Over half the patients in the molecular alteration cohorts received sitravatinib at the previously established MTD of 150 mg QD (n = 60of 113) [4]. However, based on tolerability observations during the course of the study, 120 mg QD was identified as the recommended dose, which was received by n = 53in the molecular alteration cohorts. The tolerability profile of sitravatinib is likely better with this lower dose. Furthermore, 120 mg QD was considered a clinically active dose of sitravatinib based on concentration-dependent modulation of VEGF-A and soluble-VEGF-receptor 2 in plasma samples obtained from patients before and after sitravatinib administration [4]. However, further evaluation of clinical activity according to dose would be required, along with consideration of the exposureresponse relationship for each molecular target. In addition, consideration of other molecular alterations present in the tumor may also be required, given the potential that the driver mutation may differ from the alteration selected for investigation. Sitravatinib may also have potential as combination therapy with immunotherapy, given the impact of targeting TAM receptors on immunosuppression in the tumor microenvironment [6]. However, while, combining sitravatinib with an antiprogrammed cell death protein 1 antibody (tislelizumab) demonstrated preliminary signals of antitumor activity in patients with hepatocellular and gastric cancer, clinically meaningful responses were not seen with sitravatinib plus nivolumab in patients with urothelial cancer [25,26].

# 5. Conclusion

Single-agent sitravatinib demonstrated modest clinical activity with a manageable safety profile in patients with heavily pretreated advanced tumors including NSCLC in molecularly defined cohorts (*RET* rearrangement, *MET* alterations, *CBL* alterations and *AXL* amplification). Further refining molecular alteration subtype in some target genes could identify populations in whom sitravatinib may have potential clinical utility. However, despite the basket cohort-approach to enrollment, given the low frequency of these alterations this was not feasible in the timeframe for this study. Further development of sitravatinib is not anticipated.

#### Article highlights

- Basket trials that enroll cohorts of patients with various tumor types that harbor specific molecular alterations can inform the activity of novel treatments for rare genetic drivers, helping ensure patients receive treatment that is based on the molecular signature of their disease.
- Sitravatinib (MGCD516) is an oral small molecule inhibitor that targets a spectrum of closely related receptor tyrosine kinases involved in cancer development, predominantly TAM family

(TYRO3, AXL, MERTK) and split family (VEGFR2, MET, RET, KIT) receptors.

- We report clinical activity and safety with sitravatinib in patients with advanced solid tumors that harbored genetic alterations relevant to the mechanism of action of sitravatinib, who participated in the Phase Ib basket study cohorts of the first-in-human study.
- Overall, 113 heavily pre-treated patients were enrolled; patients with tumors containing alterations in *RET* (n = 31), *CBL* (n = 31) and *MET* (n = 17) were the most frequent cohorts.
- Overall, the clinical activity of sitravatinib was modest: while most patients (68.9%) experienced a reduction in tumor volume, the majority (61.5%) had a best objective response of stable disease.
- Objective response rate was highest in patients with *RET*-rearranged NSCLC (21.1%) but this did not differ significantly from the null hypothesis (ORR  $\leq$  15%; *p* = 0.316).
- Despite the basket-cohort approach to enrollment, fewer patients than planned were enrolled across the cohorts in the timeframe of the study; this prevented further refinement of molecular alteration subtypes for which sitravatinib may have clinical utility.
- Sitravatinib had a manageable safety profile: most adverse events were mild-to-moderate in severity, with diarrhea (61.1%), fatigue (50.4%) and hypertension (46.9%) being most frequently observed.

# **Acknowledgments**

We thank the patients who participated in this study and their families.

Preliminary outcomes from these molecular alteration cohorts were reported by Bauer et al. Invest New Drugs 2022;40:990–1000 and presented at World Congress on Lung Cancer; October 2017; Yokohama, Japan (Bazhenova L et al. *CBL* mutations as potential mediators of EGFR TKI resistance effectively treated with sitravatinib. Poster P3.01–048) and European Society for Medical Oncology Annual Meeting; October 2018; Munich, Germany (Bazhenova L et al. Sitravatinib demonstrates activity in patients with novel genetic alterations that inactivate CBL. Oral presentation 408O).

# **Author contributions**

All authors contributed to data interpretation, development of the manuscript, approved the final draft for submission and take responsibility for data integrity. In addition, L Bazhenova, DW Kim, BC Cho, S Goel, R Heist, TL Werner, KD Eaton, JS Wang, S Pant, DR Adkins, C Blakely and T Bauer contributed to data collection; X Yan was responsible for data analysis and contributed to data presentation; S Neuteboom was responsible for coordinating the research and contributed to data presentation; JG Christensen and R Chao were responsible for conception and design of the study and contributed to data presentation.

# **Financial disclosure**

This study was supported by Mirati Therapeutics Inc.

# **Competing interests disclosure**

L Bazhenova: consulting fees from BioAtla, Abbvie, Gilead, Anheart, Bayer, Boehringer Ingelheim, Janssen, Bristol Myers Squibb, Daichi Sankyo, Sanofi, Elevation Oncology, Genentech, Intervenn, Merck, Neuvogen, Novocure, Pfizer, Regeneron, Summit Therapeutics, Teligene, Mirati Therapeutics, and Turning Point Therapeutics; consulting fees paid to institution from AstraZeneca and Takeda; Data Safety Monitoring Board/Advisory Board participation for Neuvogen and ORIC Pharmaceuticals; and stock/stock options for Epic Sciences. D-W Kim: research funding to institution from Alpha Biopharma, BridgeBioTherapeutics, Amgen, AstraZeneca/Medimmune, Boehringer-Ingelheim, GSK. Bristol Myers Squibb, TP Therapeutics, Chong Keun Dang, Novartis, Daiichi-Sankyo, Hanmi, IMBDx, InnoN, Janssen, Merck, Meurs, Mirati Therapeutics, MSD, ONO Pharmaceutical, Pfizer, Roche/Genentech, Takeda, Xcovery, and Yuhan; honoraria from Korean Association for Lung Cancer, Asian Thoracic Oncology Research Group, Korean Cancer Association, Korean Society of Medical Oncology, and Taiwan Lung Cancer Society; support for meeting attendance from International Association for the Study of Lung Cancer, Asian Thoracic Oncology Research Group, and Taiwan Lung Cancer Society; advisory board participation (unpaid) from Bristol Myers Squibb/ONO Pharmaceuticals, Amgen, Merck, AstraZeneca, Daiichi-Sankyo, GSK, Novelty Nobility, Janssen, MSD, Oncobix, Pfizer, SK Biopharm, and Takeda; scientific advisor for Health Insurance Review and Assessment Service, Republic of Korea; and member of Board of Directors for Korean Cancer Association, Asian Thoracic Oncology Research Group, Korean Association for Lung Cancer, and Korean Society of Medical Oncology. BC Cho: research funding from Abbvie, Dizal Pharma, Abion, Bayer, AstraZeneca, Blueprint Medicines, Boehringer Ingelheim, CJ Bioscience, Bridgebio Therapeutics, CHA Bundang Medical Center, JINTSbio, Champions Oncology, CJ Blossom Park, Lilly, Cyrus, MOGAM Institute, Dong-A ST, Genexine, GI-Cell, Illumina, GI-ImmuneOncia, Kanaph Therapeutics, Hanmi, Innovation, Interpark Bio Convergence Corp, Regeneron, Janssen, LG Chem, MSD, Novartis Oncternal, Ono, Oscotec, Therapex, Nuvalent, Vertical Bio AG, and Yuhan; consulting fees from BeiGene, Abion, Pfizer, Cyrus Therapeutics, AstraZeneca, Hanmi, Blueprint Medicines, Eli Lilly, Boehringer-Ingelheim, Onegene Biotechnology, CJ, CureLogen, GI-Cell, Guardant, HK Inno-N, Imnewrun Biosciences Inc., Bristol Myers Squibb, Janssen, Ono, Medpacto, MSD, Novartis, RandBio, Takeda, and Yuhan; honoraria from Chinese Thoracic Oncology Society, American Society of Clinical Oncology, International Association for the Study of Lung Cancer, AstraZeneca, Korean Cancer Association, European Society of Medical Oncology, Guardant, Korean Cancer Study Group, Korean Society of Medical Oncology; MSD, Novartis, Korean Society of Thyroid, Head and Neck Surgery, Roche, and Pfizer; advisory board roles for Guardant Health, Amgen, Cyrus Therapeutics, Bridgebio Therapeutics, KANAPH Therapeutic Inc, Gilead, J INTS BIO, Oscotec Inc, and Therapex; royalties from Crown Bioscience, Bio GmbH, Champions Oncology, Imagen, PearlRiver, and Roche; member of the board of directors for J INTS BIO; founder of DAAN Biotherapeutics; and stock ownership for Cyrus Therapeutics, Gencurix Inc, Bridgebio Therapeutics, Interpark Bio Convergence Corp., J INTS BIO, KANAPH Therapeutics, and TheraCanVac Inc. S Goel: research support to institution from Mirati Therapeutics Inc. R Heist: research funding to institution from Daiichi Sankyo, Abbvie, Corvus, Erasca, Lilly, Mirati Therapeutics, Agios, Mythic, Novartis, and Turning Point; and consulting fees from Abbvie, Claim Therapeutics, Amgen, Regeneron, EMD Serono, AstraZeneca, Daiichi Sankyo, Lilly, Merck, Novartis, and Sanofi. TL Werner: research support to institution from Abbvie, Mersana Therapeutics, Acrivon,

AstraZeneca, Genmab, BluePrint Medicines, GSK-Tesaro, Repare Therapeutics, Clovis Oncology, and Roche Genetech; and honoraria for advisory roles from Mersana Therapeutics. KD Eaton: research support to institution from Mirati Therapeutics Inc. JS Wang: consulting fees from Kanaph Therapeutics, honoraria from AstraZeneca and Eisai; and advisory roles for Janssen Oncology, BioNTech, Fusion Pharmaceuticals, and Stemline/Menarini. S Pant: research support to institution for Astellas, Amal Therapeutics, Bristol Myers Squibb, Arcus, Lilly, Biontech, 4D Pharma, Elicio, Framewave, Immuneering, ImmunoMET, Ipsen, Janssen, Boehringer Ingelheim, Mirati Therapeutics, NGM Pharmaceuticals, Pfizer, Xencor, Novartis, and, Zymeworks; and consulting/advisory fees from AstraZeneca, Jazz Pharmaceuticals, Alligator Bioscience, AskGene Pharma, BPGBio, Ipsen, Janssen, Boehringer Ingelheim, Nihon Medi-Physics Co, Ltd, USWorldmeds, Novartis, and Zymeworks. DR Adkins: research support to institution from Beigene, Adlai Nortye, Pfizer, Calliditas, AstraZeneca, BioAtla, Celgene/Bristol Meyers Squibb, Cue Biopharma, Gilead Sciences, Debiopharm, Boehringer Ingelheim, Eli Lilly, Epizyme, Blueprint Medicine, Genmab, Cofactor Genomics, Hookipa, ISA Pharmaceuticals, Kura Oncology, Immutep, Merck, Mirati Therapeutics, Novartis, Roche, Tizona, Natco, and Vaccinex; consulting fees for Advisory Board participation from Coherus Biosciences, Blueprint Medicine, Gilead Sciences, Boehringer Ingelheim, Cue Biopharma, Immunitas, Eisai Europe, Exelixis, Jazz Pharmaceuticals, Merck, Kura Oncology, Targimmue Therapeutics, Vaxxinex, TwoXAR, and Xilio Therapeutics; travel support for meeting attendance to institution from Natco; Chairperson for Washington University Standing Data Safety Monitoring Board; Chairperson of BJC Pharmacy and Therapeutics Committee; and member of NCI Head and Neck Cancer Steering Committee. CM Blakely: research support from AstraZeneca, Puma, Mirati Therapeutics Inc., Genentech, Novartis, Pfizer, and Takeda; and consulting fees from Bayer, Gilead, Bristol Meyers Squibb, and Janssen. X Yan: employment, stock ownership and support for meeting attendance for Mirati Therapeutics Inc. S Neuteboom: prior employment and stock ownership for Mirati Therapeutics Inc. JG Christensen: employment and stock ownership for Mirati Therapeutics Inc., and Fiduciary Officer for Mirati Therapeutics Inc. and Boundless Bio. R Chao: employment, stock ownership and support for meeting attendance for Mirati Therapeutics Inc. T Bauer: consulting fees from Pfizer, Bayer, Lilly, and Sanofi; honoraria from Lilly, Bayer, and Pfizer; and participation on an Advisory Board/Data Safety Monitoring Board for Pfizer. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

# Writing disclosure

Medical writing services were provided in accordance with Good Publication Practice guidelines (http://www.ismpp.org) by Siân Marshall (SIANTIFIX Ltd, Cambridge, UK). Medical writing support was funded by Mirati Therapeutics, Inc.

# **Ethical conduct of research**

The study was conducted in accordance with the Declaration of Helsinki and Good Clinical Practice guidelines, defined by the

International Council for Harmonisation, and was approved by the institutional review board at each participating site. Western Institutional Review Board served as the central institutional review board in conjunction with 23 local institutional review boards. All participants provided written, informed consent.

# **Data availability**

Mirati Therapeutics Inc. (a Bristol Myers Squibb company) is committed to patient care, advancing scientific understanding, and enabling the scientific community to learn from and build upon our research. We will honor legitimate requests from qualified researchers and investigators for our clinical trial data in order to conduct methodologically sound research. We will share clinical trial data, as well as study protocols, clinical study reports and statistical analysis plans for this study. Sharing is subject to protecting patient privacy and respect for the patient's informed consent. Data will generally be made available for specific requests approximately 2 years after clinical trial completion. For additional information on data sharing collaborations with Mirati Therapeutics, please email medinfo@mirati.com.

# ORCID

Lyudmila Bazhenova () https://orcid.org/0000-0001-8764-4359

# References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1. Du Z, Lovly CM. Mechanisms of receptor tyrosine kinase activation in cancer. Mol Cancer. 2018;17(1):58. doi:10.118 6/s12943-018-0782-4
- Yamaoka T, Kusumoto S, Ando K, et al. Receptor tyrosine kinase-targeted cancer therapy. Int J Mol Sci. 2018;19(11):3491. doi:10.3390/ijms19113491
- 3. Pottier C, Fresnais M, Gilon M, et al. Tyrosine kinase inhibitors in cancer: breakthrough and challenges of targeted therapy. Cancers (Basel). 2020;12(3):731. doi:10 .3390/cancers12030731
- Bauer T, Cho BC, Heist R, et al. First-in-human Phase I/Ib study to evaluate sitravatinib in patients with advanced solid tumors. Invest New Drugs. 2022;40(5):990–1000. doi:10.1007/s10637-022-01274-y

•• Describes the design of this study and pharmacokinetic/pharmacodynamic, safety and efficacy findings at primary data cutoff.

 Patwardhan PP, Ivy KS, Musi E, et al. Significant blockade of multiple receptor tyrosine kinases by MGCD516 (sitravatinib), a novel small molecule inhibitor, shows potent anti-tumor activity in preclinical models of sarcoma. Oncotarget. 2016;7(4):4093–5109. doi:10.18632/oncotar get.6547

# •• Demonstrates potent antitumor activity of sitravatinib in sarcoma xenograft models.

 Du W, Huang H, Sorrelle N, et al. Sitravatinib potentiates immune checkpoint blockade in refractory cancer models. JCI Insight. 2018;3(21):e124184. doi:10.1172/jci.insigh t.124184

• Demonstrates the impact of sitravatinib on the tumor microenvironment and antitumor activity in xenograft models.

- 7. Zhang Y, Wang P, Wang Y, et al. Sitravatinib as a potent FLT3 inhibitor can overcome gilteritinib resistance in acute myeloid leukemia. Biomark Res. 2023;11(1):8. doi:10.1186/s40364-022-00447-4
- Park JJH, Hsu G, Siden EG, et al. An overview of precision oncology basket and umbrella trials for clinicians. CA Cancer J Clin. 2020;70(2):125–137. doi:10.3322/caac.216 00

# • Describes key characteristics and considerations for basket and umbrella clinical trials.

9. Haslam A, Olivier T, Tuia J, et al. Umbrella review of basket trials testing a drug in tumors with actionable genetic biomarkers. BMC Cancer. 2023;23(1):46. doi:10.1186/s128 85-022-10421-w

# • Provides insight on outcomes from basket studies of oncology drugs.

10. Mansinho A, Fernandes RM, Carneiro AV. Histologyagnostic drugs: a paradigm shift - a narrative review. Adv Ther. 2023;40(4):1379–1392. doi:10.1007/s12325-022 -02362-4

• Reviews insights and challenges for histologyagnostic, molecularly targeted approaches for clinical trials.

- Drilon A, Laetsch TW, Kummar S, et al. Efficacy of larotrectinib in *TRK* fusion-positive cancers in adults and children. N Engl J Med. 2018;378(8):731–739. doi:10.105 6/NEJMoa1714448
- 12. Disel U, Madison R, Abhishek K, et al. The pan-cancer landscape of co-amplification of the tyrosine kinases KIT, KDR, and PDGFRA. Oncologist. 2020;25(1):e39–e47. doi:10.1634/theoncologist.2018-0528
- Tang R, Langdon WY, Zhang J. Negative regulation of receptor tyrosine kinases by ubiquitination: key roles of the Cbl family of E3 ubiquitin ligases. Front Endocrinol (Lausanne). 2022;13:971162. doi:10.3389/fendo.2022.971 162
- 14. Tan YC, Mirzapoiazova T, Won BM, et al. Differential responsiveness of MET inhibition in non-small-cell lung cancer with altered CBL. Sci Rep. 2017;7(1):9192. doi:10.1 038/s41598-017-09078-4
- Scott RP, Eketjall S, Aineskog H, et al. Distinct turnover of alternatively spliced isoforms of the RET kinase receptor mediated by differential recruitment of the Cbl ubiquitin ligase. J Biol Chem. 2005;280(14):13442–13449. doi:10.1 074/jbc.M500507200
- Bradford D, Larkins E, Mushti SL, et al. FDA approval summary: selpercatinib for the treatment of lung and thyroid cancers with RET gene mutations or fusions. Clin Cancer Res. 2021;27(8):2130–2135. doi:10.1158/1078-043 2.CCR-20-3558

- 17. Kim J, Bradford D, Larkins E, et al. FDA approval summary: pralsetinib for the treatment of lung and thyroid cancers with RET gene mutations or fusions. Clin Cancer Res. 2021;27(20):5452–5456. doi:10.1158/1078-0 432.CCR-21-0967
- Desai A, Cuellar S. The current landscape for METex14 skipping mutations in non-small cell lung cancer. J Adv Pract Oncol. 2022;13(5):539–544. doi:10.6004/jadpro.202 2.13.5.8
- 19. Le X, Sakai H, Felip E, et al. Tepotinib efficacy and safety in patients with MET exon 14 skipping NSCLC: outcomes in patient subgroups from the VISION Study with relevance for clinical practice. Clin Cancer Res. 2022;28(6):1117–1126. doi:10.1158/1078-0432.CCR-21-2733
- Wolf J, Seto T, Han JY, et al. Capmatinib in MET exon 14mutated or MET-amplified non-small-cell lung cancer. N Engl J Med. 2020;383(10):944–957. doi:10.1056/NEJMoa 2002787
- 21. Tang Y, Zang H, Wen Q, et al. AXL in cancer: a modulator of drug resistance and therapeutic target. J Exp Clin Cancer Res. 2023;42(1):148. doi:10.1186/s13046-023-027 26-w
- 22. Oliva M, Chepeha D, Araujo DV, et al. Antitumor immune effects of preoperative sitravatinib and nivolumab in oral cavity cancer: SNOW window-of-opportunity study. J Immunother Cancer. 2021;9(10):e003476. doi:10.1136/ji tc-2021-003476
- 23. Msaouel P, Goswami S, Thall PF, et al. A Phase I– Il trial of sitravatinib and nivolumab in clear cell renal cell carcinoma following progression on antiangiogenic therapy. Sci Transl Med. 2022;14(641):eabm6420. doi:10.1 126/scitranslmed.abm6420
- 24. Karam JA, Msaouel P, Haymaker CL, et al. Phase II trial of neoadjuvant sitravatinib plus nivolumab in patients undergoing nephrectomy for locally advanced clear cell renal cell carcinoma. Nat Commun. 2023;14(1):2684. doi:10.1038/s41467-023-38342-7
- 25. Msaouel P, Sweis RF, Bupathi M, et al. A Phase II study of sitravatinib in combination with nivolumab in patients with advanced or metastatic urothelial carcinoma. Eur Urol Oncol. 2024;7(4):933–943. doi:10.1016/j.euo.2023.1 2.001
- 26. Li J, Bai Y, Chen Z, et al. SAFFRON-104: a Phase Ib/II study of sitravatinib alone or with tislelizumab in advanced hepatocellular carcinoma and gastric cancer/gastroesophageal junction cancer. Cancer Immunol Immunother. 2024;73(11):219. doi:10.1007/s00262-024-0 3806-2